Author’s response to reviews

Title: A right to health compliant global tiered pricing framework?

Authors:

Gorik Ooms (gooms@itg.be)
Lisa Forman (lisa.forman@utoronto.ca)
Owain D Williams (odw@aber.ac.uk)
Peter S Hill (p.hill@sph.uq.edu.au)

Version: 2 Date: 9 September 2014

Author’s response to reviews:

This paper is intended for the debate section, and the analysis uses only data that have been gathered and reported by others. We did not ask for approval by an ethics committee.